\xname{H2-Air11}
\W\subsubsection{Critical diameter vs equivalence ratio; H2-Air}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Air11.ps,width=6in}}
{\htmlimage{\plotdir/H2-Air11.gif}}
\T\caption{Critical diameter vs equivalence ratio; H2-Air}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Air11.ps}

\T mk23a - Table~\ref{mk23a}~\cite[Guirao (1982)]{c_guir82}
\W\xlink{mk23a.txt}{\rawdatadir/mk23a.txt}
\W Guirao (1982) \cite{c_guir82}
T=293 K, P=101.3 kPa

\T mk23b - Table~\ref{mk23b}~\cite[Guirao (1982)]{c_guir82}
\W\xlink{mk23b.txt}{\rawdatadir/mk23b.txt}
\W Guirao (1982) \cite{c_guir82}
T=293 K, P=101.3 kPa

\T\clearpage

\xname{H2-Ox23}
\W\subsubsection{Critical diameter vs percent N2; H2-O2-N2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Ox23.ps,width=6in}}
{\htmlimage{\plotdir/H2-Ox23.gif}}
\T\caption{Critical diameter vs percent N2; H2-O2-N2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Ox23.ps}

\T at72c - Table~\ref{at72c}~\cite[Matsui (1979)]{h_mats79}
\W\xlink{at72c.txt}{\rawdatadir/at72c.txt}
\W Matsui (1979) \cite{h_mats79}
P=101.3 kPa, ER=1

\T at72a - Table~\ref{at72a}~\cite[Knystautas (1982)]{r_knys82}
\W\xlink{at72a.txt}{\rawdatadir/at72a.txt}
\W Knystautas (1982) \cite{r_knys82}
P=101.3 kPa, ER=1

\T\clearpage

\xname{H2-Ox24}
\W\subsubsection{Critical diameter vs equivalence ratio; H2-O2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Ox24.ps,width=6in}}
{\htmlimage{\plotdir/H2-Ox24.gif}}
\T\caption{Critical diameter vs equivalence ratio; H2-O2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Ox24.ps}

\T at43 - Table~\ref{at43}~\cite[Plaster (1991)]{m_plas91}
\W\xlink{at43.txt}{\rawdatadir/at43.txt}
\W Plaster (1991) \cite{m_plas91}
T=100 K, P=100 kPa

\T at56a - Table~\ref{at56a}~\cite[Makris (1994)]{a_makr94}
\W\xlink{at56a.txt}{\rawdatadir/at56a.txt}
\W Makris (1994) \cite{a_makr94}
T=293 K, P=101.3 kPa

\T\clearpage

\xname{H2-Ox25}
\W\subsubsection{Critical diameter vs initial pressure; H2-O2-Additive}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Ox25.ps,width=6in}}
{\htmlimage{\plotdir/H2-Ox25.gif}}
\T\caption{Critical diameter vs initial pressure; H2-O2-Additive}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Ox25.ps}

\T at75a - Table~\ref{at75a}~\cite[Matsui (1979)]{h_mats79}
\W\xlink{at75a.txt}{\rawdatadir/at75a.txt}
\W Matsui (1979) \cite{h_mats79}
T=293 K, ER=1

\T at75b - Table~\ref{at75b}~\cite[Moen (1985)]{i_moen85}
\W\xlink{at75b.txt}{\rawdatadir/at75b.txt}
\W Moen (1985) \cite{i_moen85}
T=293 K, ER=1

\T at75c - Table~\ref{at75c}~\cite[Moen (1985)]{i_moen85}
\W\xlink{at75c.txt}{\rawdatadir/at75c.txt}
\W Moen (1985) \cite{i_moen85}
T=293 K, ER=1, 5\% CO2

\T at75d - Table~\ref{at75d}~\cite[Moen (1985)]{i_moen85}
\W\xlink{at75d.txt}{\rawdatadir/at75d.txt}
\W Moen (1985) \cite{i_moen85}
T=293 K, ER=1, 5\% CF3Br

\T\clearpage

\xname{CH4-Ox6}
\W\subsubsection{Critical diameter vs initial pressure (stoichiometric); CH4-O2-N2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/CH4-Ox6.ps,width=6in}}
{\htmlimage{\plotdir/CH4-Ox6.gif}}
\T\caption{Critical diameter vs initial pressure (stoichiometric); CH4-O2-N2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/CH4-Ox6.ps}

\T at147a - Table~\ref{at147a}~\cite[Knystautas (1982)]{r_knys82}
\W\xlink{at147a.txt}{\rawdatadir/at147a.txt}
\W Knystautas (1982) \cite{r_knys82}
ER=1, 36.2\% N2

\T att2 - Table~\ref{att2}~\cite[Zeldovich (1956)]{y_zeld56}
\W\xlink{att2.txt}{\rawdatadir/att2.txt}
\W Zeldovich (1956) \cite{y_zeld56}
ER=1

\T mk137k - Table~\ref{mk137k}~\cite[Matsui (1979)]{h_mats79}
\W\xlink{mk137k.txt}{\rawdatadir/mk137k.txt}
\W Matsui (1979) \cite{h_mats79}
ER=1

\T\clearpage

\xname{CH4-Ox7}
\W\subsubsection{Critical diameter vs initial pressure; CH4-O2-N2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/CH4-Ox7.ps,width=6in}}
{\htmlimage{\plotdir/CH4-Ox7.gif}}
\T\caption{Critical diameter vs initial pressure; CH4-O2-N2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/CH4-Ox7.ps}

\T at161a - Table~\ref{at161a}~\cite[Bauer (1984)]{p_baue84}
\W\xlink{at161a.txt}{\rawdatadir/at161a.txt}
\W Bauer (1984) \cite{p_baue84}
ER=1.08, 31.9\% N2

\T at161b - Table~\ref{at161b}~\cite[Bauer (1984)]{p_baue84}
\W\xlink{at161b.txt}{\rawdatadir/at161b.txt}
\W Bauer (1984) \cite{p_baue84}
ER=1.09, 54.9\% N2

\T at161c - Table~\ref{at161c}~\cite[Bauer (1984)]{p_baue84}
\W\xlink{at161c.txt}{\rawdatadir/at161c.txt}
\W Bauer (1984) \cite{p_baue84}
ER=1.15, 64.3\% N2

\T at161d - Table~\ref{at161d}~\cite[Bauer (1984)]{p_baue84}
\W\xlink{at161d.txt}{\rawdatadir/at161d.txt}
\W Bauer (1984) \cite{p_baue84}
ER=1.09, 67.7\% N2

\T at147b - Table~\ref{at147b}~\cite[Matsui (1979)]{h_mats79}
\W\xlink{at147b.txt}{\rawdatadir/at147b.txt}
\W Matsui (1979) \cite{h_mats79}
ER=1.3

\T at147c - Table~\ref{at147c}~\cite[Matsui (1979)]{h_mats79}
\W\xlink{at147c.txt}{\rawdatadir/at147c.txt}
\W Matsui (1979) \cite{h_mats79}
ER=1.3, 23\% N2

\T at147d - Table~\ref{at147d}~\cite[Matsui (1979)]{h_mats79}
\W\xlink{at147d.txt}{\rawdatadir/at147d.txt}
\W Matsui (1979) \cite{h_mats79}
ER=1.3, 50\% N2

\T\clearpage

\xname{C2H2-Ox7}
\W\subsubsection{Critical diameter vs percent diluent; C2H2-O2-N2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H2-Ox7.ps,width=6in}}
{\htmlimage{\plotdir/C2H2-Ox7.gif}}
\T\caption{Critical diameter vs percent diluent; C2H2-O2-N2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H2-Ox7.ps}

\T at71a - Table~\ref{at71a}~\cite[Knystautas (1982)]{r_knys82}
\W\xlink{at71a.txt}{\rawdatadir/at71a.txt}
\W Knystautas (1982) \cite{r_knys82}
ER=1

\T at66a - Table~\ref{at66a}~\cite[Matsui (1979)]{h_mats79}
\W\xlink{at66a.txt}{\rawdatadir/at66a.txt}
\W Matsui (1979) \cite{h_mats79}
ER=1

\T at66b - Table~\ref{at66b}~\cite[Matsui (1979)]{h_mats79}
\W\xlink{at66b.txt}{\rawdatadir/at66b.txt}
\W Matsui (1979) \cite{h_mats79}
ER=1.67

\T mk127 - Table~\ref{mk127}~\cite[Zeldovich (1956)]{y_zeld56}
\W\xlink{mk127.txt}{\rawdatadir/mk127.txt}
\W Zeldovich (1956) \cite{y_zeld56}
ER=1

\T\clearpage

\xname{C2H2-Air4}
\W\subsubsection{Critical diameter vs equivalence ratio; C2H2-Air}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H2-Air4.ps,width=6in}}
{\htmlimage{\plotdir/C2H2-Air4.gif}}
\T\caption{Critical diameter vs equivalence ratio; C2H2-Air}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H2-Air4.ps}

\T at22a1 - Table~\ref{at22a1}~\cite[Moen (1984)]{i_moen84}
\W\xlink{at22a1.txt}{\rawdatadir/at22a1.txt}
\W Moen (1984) \cite{i_moen84}


\T at20a - Table~\ref{at20a}~\cite[Rinnan (1982)]{a_rinn82}
\W\xlink{at20a.txt}{\rawdatadir/at20a.txt}
\W Rinnan (1982) \cite{a_rinn82}


\T at22a4 - Table~\ref{at22a4}~\cite[Matsui (1979)]{h_mats79}
\W\xlink{at22a4.txt}{\rawdatadir/at22a4.txt}
\W Matsui (1979) \cite{h_mats79}


\T\clearpage

\xname{C2H4-Air4}
\W\subsubsection{Critical diameter vs equivalence ratio; C2H4-Air-Inhibitor}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H4-Air4.ps,width=6in}}
{\htmlimage{\plotdir/C2H4-Air4.gif}}
\T\caption{Critical diameter vs equivalence ratio; C2H4-Air-Inhibitor}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H4-Air4.ps}

\T at31c - Table~\ref{at31c}~\cite[Rinnan (1982)]{a_rinn82}
\W\xlink{at31c.txt}{\rawdatadir/at31c.txt}
\W Rinnan (1982) \cite{a_rinn82}


\T at22b - Table~\ref{at22b}~\cite[Moen (1984)]{i_moen84}
\W\xlink{at22b.txt}{\rawdatadir/at22b.txt}
\W Moen (1984) \cite{i_moen84}


\T at22c - Table~\ref{at22c}~\cite[Moen (1982)]{i_moen82}
\W\xlink{at22c.txt}{\rawdatadir/at22c.txt}
\W Moen (1982) \cite{i_moen82}


\T at77a - Table~\ref{at77a}~\cite[Moen (1985)]{i_moen85}
\W\xlink{at77a.txt}{\rawdatadir/at77a.txt}
\W Moen (1985) \cite{i_moen85}
1.5\% CF4

\T at77b - Table~\ref{at77b}~\cite[Moen (1985)]{i_moen85}
\W\xlink{at77b.txt}{\rawdatadir/at77b.txt}
\W Moen (1985) \cite{i_moen85}
1.5\% CF3Br

\T at77c - Table~\ref{at77c}~\cite[Moen (1985)]{i_moen85}
\W\xlink{at77c.txt}{\rawdatadir/at77c.txt}
\W Moen (1985) \cite{i_moen85}
3\% CF3Br

\T at77d - Table~\ref{at77d}~\cite[Moen (1985)]{i_moen85}
\W\xlink{at77d.txt}{\rawdatadir/at77d.txt}
\W Moen (1985) \cite{i_moen85}
3\% CO2

\T\clearpage

\xname{C2H4-Ox4}
\W\subsubsection{Critical diameter vs equivalence ratio; C2H4-O2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H4-Ox4.ps,width=6in}}
{\htmlimage{\plotdir/C2H4-Ox4.gif}}
\T\caption{Critical diameter vs equivalence ratio; C2H4-O2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H4-Ox4.ps}

\T at56c - Table~\ref{at56c}~\cite[Makris (1994)]{a_makr94}
\W\xlink{at56c.txt}{\rawdatadir/at56c.txt}
\W Makris (1994) \cite{a_makr94}


\T at56d - Table~\ref{at56d}~\cite[Matsui (1979)]{h_mats79}
\W\xlink{at56d.txt}{\rawdatadir/at56d.txt}
\W Matsui (1979) \cite{h_mats79}


\T at80b - Table~\ref{at80b}~\cite[Matsui (1979)]{h_mats79}
\W\xlink{at80b.txt}{\rawdatadir/at80b.txt}
\W Matsui (1979) \cite{h_mats79}


\T at80c - Table~\ref{at80c}~\cite[Matsui (1979)]{h_mats79}
\W\xlink{at80c.txt}{\rawdatadir/at80c.txt}
\W Matsui (1979) \cite{h_mats79}


\T\clearpage

\xname{C2H4-Ox5}
\W\subsubsection{Critical diameter vs initial pressure; C2H4-O2-N2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H4-Ox5.ps,width=6in}}
{\htmlimage{\plotdir/C2H4-Ox5.gif}}
\T\caption{Critical diameter vs initial pressure; C2H4-O2-N2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H4-Ox5.ps}

\T at137g - Table~\ref{at137g}~\cite[Matsui (1979)]{h_mats79}
\W\xlink{at137g.txt}{\rawdatadir/at137g.txt}
\W Matsui (1979) \cite{h_mats79}


\T at162a - Table~\ref{at162a}~\cite[Bauer (1984)]{p_baue84}
\W\xlink{at162a.txt}{\rawdatadir/at162a.txt}
\W Bauer (1984) \cite{p_baue84}
71.4\% N2

\T at162b - Table~\ref{at162b}~\cite[Bauer (1984)]{p_baue84}
\W\xlink{at162b.txt}{\rawdatadir/at162b.txt}
\W Bauer (1984) \cite{p_baue84}
67.2\% N2

\T at162c - Table~\ref{at162c}~\cite[Bauer (1984)]{p_baue84}
\W\xlink{at162c.txt}{\rawdatadir/at162c.txt}
\W Bauer (1984) \cite{p_baue84}
60.1\% N2

\T\clearpage

\xname{C2H4-Ox6}
\W\subsubsection{Critical diameter vs percent diluent; C2H4-O2-N2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H4-Ox6.ps,width=6in}}
{\htmlimage{\plotdir/C2H4-Ox6.gif}}
\T\caption{Critical diameter vs percent diluent; C2H4-O2-N2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H4-Ox6.ps}

\T at200c - Table~\ref{at200c}~\cite[Moen (1981)]{i_moen81}
\W\xlink{at200c.txt}{\rawdatadir/at200c.txt}
\W Moen (1981) \cite{i_moen81}
ER=1

\T at66c - Table~\ref{at66c}~\cite[Matsui (1979)]{h_mats79}
\W\xlink{at66c.txt}{\rawdatadir/at66c.txt}
\W Matsui (1979) \cite{h_mats79}
ER=1.5

\T at163a - Table~\ref{at163a}~\cite[Bauer (1984)]{p_baue84}
\W\xlink{at163a.txt}{\rawdatadir/at163a.txt}
\W Bauer (1984) \cite{p_baue84}
ER=1.05

\T mk142a - Table~\ref{mk142a}~\cite[Knystautas (1982)]{r_knys82}
\W\xlink{mk142a.txt}{\rawdatadir/mk142a.txt}
\W Knystautas (1982) \cite{r_knys82}
ER=1

\T\clearpage

\xname{C2H6-Air2}
\W\subsubsection{Critical diameter vs equivalence ratio; C2H6-Air}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H6-Air2.ps,width=6in}}
{\htmlimage{\plotdir/C2H6-Air2.gif}}
\T\caption{Critical diameter vs equivalence ratio; C2H6-Air}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H6-Air2.ps}

\T mk22e - Table~\ref{mk22e}~\cite[Moen (1984)]{i_moen84}
\W\xlink{mk22e.txt}{\rawdatadir/mk22e.txt}
\W Moen (1984) \cite{i_moen84}


\T\clearpage

\xname{HC-Ox1}
\W\subsubsection{Critical diameter vs percent diluent; 2C2H6+5O2+nN2, 2C3H8+7O2+nN2, C3H6+3O2+nN2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/HC-Ox1.ps,width=6in}}
{\htmlimage{\plotdir/HC-Ox1.gif}}
\T\caption{Critical diameter vs percent diluent; 2C2H6+5O2+nN2, 2C3H8+7O2+nN2, C3H6+3O2+nN2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/HC-Ox1.ps}

\T at66d - Table~\ref{at66d}~\cite[Matsui (1979)]{h_mats79}
\W\xlink{at66d.txt}{\rawdatadir/at66d.txt}
\W Matsui (1979) \cite{h_mats79}
Fuel=C2H6, Oxidizer=O2

\T at66e - Table~\ref{at66e}~\cite[Matsui (1979)]{h_mats79}
\W\xlink{at66e.txt}{\rawdatadir/at66e.txt}
\W Matsui (1979) \cite{h_mats79}
Fuel=C3H8, Oxidizer=O2

\T at66f - Table~\ref{at66f}~\cite[Matsui (1979)]{h_mats79}
\W\xlink{at66f.txt}{\rawdatadir/at66f.txt}
\W Matsui (1979) \cite{h_mats79}
Fuel=C3H6, Oxidizer=O2

\T at143a - Table~\ref{at143a}~\cite[Knystautas (1982)]{r_knys82}
\W\xlink{at143a.txt}{\rawdatadir/at143a.txt}
\W Knystautas (1982) \cite{r_knys82}
Fuel=C2H6, Oxidizer=O2

\T\clearpage

\xname{HC-Ox2}
\W\subsubsection{Critical diameter vs initial pressure; C2H6-O2, C3H8-O2, C3H6-O2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/HC-Ox2.ps,width=6in}}
{\htmlimage{\plotdir/HC-Ox2.gif}}
\T\caption{Critical diameter vs initial pressure; C2H6-O2, C3H8-O2, C3H6-O2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/HC-Ox2.ps}

\T at137h - Table~\ref{at137h}~\cite[Matsui (1979)]{h_mats79}
\W\xlink{at137h.txt}{\rawdatadir/at137h.txt}
\W Matsui (1979) \cite{h_mats79}
Fuel=C2H6

\T at137i - Table~\ref{at137i}~\cite[Matsui (1979)]{h_mats79}
\W\xlink{at137i.txt}{\rawdatadir/at137i.txt}
\W Matsui (1979) \cite{h_mats79}
Fuel=C3H8

\T at137j - Table~\ref{at137j}~\cite[Matsui (1979)]{h_mats79}
\W\xlink{at137j.txt}{\rawdatadir/at137j.txt}
\W Matsui (1979) \cite{h_mats79}
Fuel=C3H6

\T\clearpage

